Author’s response to reviews

Title: Early dissemination of bevacizumab for advanced colorectal cancer: a prospective cohort study

Authors:

S. Yousuf Zafar (yousuf.zafar@duke.edu)
Jennifer L Malin (Jennifer.Malin@va.gov)
Steven C Grambow (steven.grambow@duke.edu)
David H Abbott (David.Abbott@va.gov)
Deborah Schrag (Deb_Schrag@dfci.harvard.edu)
Jane T Kolimaga (kolim001@mc.duke.edu)
Leah L Zullig (Leah.Zullig@va.gov)
Jane C Weeks (Jane_Weeks@dfci.harvard.edu)
Mona N Fouad (mfouad@uab.edu)
John Z Ayanian (ayanian@hcp.med.harvard.edu)
Robert Wallace (robert-wallace@uiowa.edu)
Katherine L Kahn (kahn@rand.org)
Patricia A Ganz (pganz@ucla.edu)
Paul Catalano (pcata@iimmy.harvard.edu)
Dee W West (Dee.West@cpic.org)
Dawn Provenzale (dawn.provenzale@duke.edu)

Version: 3  Date: 8 August 2011

Author’s response to reviews:

Please note that all major revisions and most minor revisions have been completed including:

- Removal of shading in tables
- Referral of additional files as such
- Addition of "Additional files" section at the end of the manuscript
- Addition of country to affiliations
- Removal of highlighting and tracked changes
- Removal of vertical lines from tables
- Removal of titles from figures